Author’s response to reviews

Title: L-Citrulline increases hepatic sensitivity to insulin by reducing the phosphorylation of serine 1101 in insulin receptor substrate-1

Authors:

Hisae Yoshitomi (hisa4403@mukogawa-u.ac.jp)
Maki Momoo (peachmomo6@hotmail.com)
Xiao Ma (ma.xiao.edu@gmail.com)
Yewei Huang (huangyewei2011@gmail.com)
Shiori Suguro (suguro@prochemi.ne.jp)
Yoshie Yamagishi (yamagishi@prochemi.ne.jp)
Ming Gao (gaoming@mukogawa-u.ac.jp)

Version: 6 Date: 28 May 2015

Author’s response to reviews: see over
May 28, 2015

MS: 3930302531476936

L-Citrulline increases hepatic sensitivity to insulin by reducing the phosphorylation of serine 1101 in insulin receptor substrate-1

Hisae Yoshitomi, Maki Momoo, Xiao Ma, Yewei Huang, Shiori Suguro, Yoshie Yamagishi and Ming Gao

Dear Editor,

Thank you very much for your letter dated May 27, 2015 together with the reviewers’ comments.

Based on the editorial point, we include details in the Methods section on the source of the L-Citrulline used in our study, page 6, line97.

We hope that our revised manuscript is now acceptable for publication in BMC Complementary and Alternative medicine.

With kindest regards,

Sincerely yours,

Hisae Yoshitomi and Ming Gao

School of Pharmaceutical Sciences, Mukogawa Women’s University, 11-68 Koshien Kyuban-cho, Nishinomiya, Hyogo 663-8179, Japan.

Email: hisa4403@mukogawa-u.ac.jp, gaoming@mukogawa-u.ac.jp